Author's response to reviews

Title: Cohort study ON Neuroimaging, Etiology and Cognitive consequences of Transient neurological attacks (CONNECT): study rationale and protocol

Authors:

Frank G van Rooij (frank.vanrooij@radboudumc.nl)
Anil M Tuladhar (anil.tuladhar@radboudumc.nl)
Roy PC Kessels (roy.kjessels@radboudumc.nl)
Sarah E Vermeer (svermeer@rijnstate.nl)
Bozena M Góraj (bozena.goraj@radboudumc.nl)
Peter J Koudstaal (p.j.koudstaal@erasmusmc.nl)
David G Norris (david.norris@donders.ru.nl)
Frank-Erik de Leeuw (frankerik.deleeuw@radboudumc.nl)
Ewoud J van Dijk (ewoud.vandijk@radboudumc.nl)

Version: 2 Date: 12 February 2015

Author's response to reviews: see over
Dear Dr. Shipley,

Please find enclosed our revised manuscript entitled “Cohort study ON Neuroimaging, Etiology and Cognitive consequences of Transient neurological attacks (CONNECT): study rationale and protocol”.

As requested we have made the following changes:

- We have added line numbering throughout the manuscript.
- We have added an Authors’ Contributions section in which we specify the contributions each author has made to the study and manuscript, in line with the suggested format.
- We have added an Acknowledgements section in which we mention the source of funding and specify the role of the funding body.
- We have removed the additional files (proofs of funding and ethics approval).

We hope that after these changes you will find our manuscript suitable for publication in BMC Neurology.

On behalf of all authors,

Yours sincerely,

Frank G. van Rooij, MD
Neurology Resident
frank.vanrooij@radboudumc.nl